Author's response to reviews

Title: Tamoxifen and the Raloxifene analog LY117018: their effects on arachidonic acid release from cells in culture and on prostaglandin I2 production by rat liver cells

Authors:

Lawrence Levine (llevine@brandeis.edu)

Version: 3 Date: 10 August 2004

Author's response to reviews: see over
Date: 8/10/04

To: Matt Hodgkinson
Assistant Editor
BMC Journals
Email: editorial@biomedcentral.com

Dear Sir,

We are submitting the revised manuscript 1521161402309337 “Tamoxifen and the Rafoxifene analog LY117018: their effects on arachidonic acid release from cells in culture and on prostaglandin I_2 production by rat liver cells”.

We have made the format changes suggested in your e-mail dated 8/5/04. We are submitting the figures in color.

Sincerely,

Lawrence Levine
Professor Emeritus of Biochemistry
Brandeis University